Author's response to reviews

Title: Successive influenza virus infection and Streptococcus pneumoniae stimulation alter human dendritic cell function

Authors:

Yuet Wu (zacharywu03@yahoo.com.hk)
Huawei Mao (hwmau@hku.hk)
Man-To Ling (vincentling2002@gmail.com)
Kin-Hung Chow (khchowb@hkusua.hku.hk)
Pak-Leung Ho (plho@hkucc.hku.hk)
Wenwei Tu (wwtu@hkucc.hku.hk)
Yu-lung Lau (lauylung@hku.hk)

Version: 4 Date: 19 July 2011

Author's response to reviews:

1. Highlighting has been removed from the manuscript
2. The term "et al" has been removed from the reference list, and the name of the authors has been added
3. On page 20, a new sentence "However, the present findings are limited to in vitro study, and future in vivo experiments would help better understand how the innate immune cells may differentially respond to the co-infection" has been added, as in response to the comments from reviewer 3
4. The word "pneumococcus" has been replaced by "pneumococci" in some places